Author’s response to reviews

Title: Mothers' Health Care Seeking Behaviour and Associated Factors for Common Childhood Illnesses, Northwest Ethiopia: Community Based Cross-sectional Study

Authors:
Muluye Molla simeneh (mulumola6@gmail.com)
Mezgebu Yitayal Mengistu (mezgebuy@gmail.com)
Abebaw Addis Gelagay (abebaw.addis@gmail.com)
Mulugeta Tesfa Gebeyehu (mulutesfa.g@gmail.com)

Version: 3 Date: 25 Aug 2018

Author’s response to reviews:

Date: August 25, 2018

To BMC Health Services Research Editorial Office

We submitted a manuscript entitled 'Mothers' Health Care Seeking Behavior and Associated Factors for Common Childhood Illnesses, Northwest Ethiopia: Community Based Cross-sectional Study' to your journal for publication and you sent reviewers' comment. We would like acknowledge editors and reviewers for valuable comments. The authors tried to address all the comments and suggestions given and we have given a point-by-point response for each reviews' and editor's comments. So, the authors submitted the point by point responses, Clean manuscript, the track change, and edited tables and figures. The authors declared that they have no any conflict of interest.

With kind regards,

Muluye Molla simeneh
Principal investigator